Author's response to reviews

**Title:** Down-regulation of the expression of CCAAT/enhancer binding protein alpha gene in cervical squamous cell carcinoma

**Authors:**

Zemin Pan (panteacher89@sina.com)
Weinan Zheng (rwz77666@163.com)
Jinli Zhang (zhangjinli@shzu.edu.cn)
Rui Gao (gaorui_shz@sina.com)
Dongmei Li (lidong_abc@126.com)
Xiaoqing Guo (gxqj66@sina.com)
Hu Han (49166070@qq.com)
Feng Li (lifeng7855@yahoo.com.cn)
Shen Qu (qushen@mails.tjmu.edu.cn)
Renfu Shao (rshao@usc.edu.au)

**Version:** 5  
**Date:** 10 April 2014

**Author's response to reviews:** see over
- Please could you reword the sentence, 'informed consent in written' to 'written informed consent was obtained from .........'.
   Answer: We have correct this sentence in our manuscript.

- Please could you mention where you obtained the HeLa cells from, in the methods section of your manuscript.
   Answer:
   HeLa cells were initially obtained from ATCC(American Type Culture Collection, Manassas, VA) and maintained in our lab.